Author’s response to reviews

Title: Magnesium supplementation and preeclampsia in low-income pregnant women – a randomized double-blind clinical trial

Authors:

Carla Araujo (carla.leal@imip.org.br)
Larissa Oliveira (larissasouzaoliveira@hotmail.com)
Isabela Gusmao (isabelagusmao@hotmail.com)
Angélica Guimaraes (angel.guimaraes@bol.com.br)
Moranna Ribeiro (moranna.ribeiro@hotmail.com)
J.G. Guilherme Bezerra Alves (joaoguilherme@imip.org.br)

Version: 3 Date: 11 Mar 2020

Author’s response to reviews:

Response Letter

We note that the current submission contains some textual overlap with other previously published works, in particular:
https://journals.plos.org/plosone/article?id=10.1371%2Fjournal.pone.0227497
Answer: We are very sorry for this. We hope all textual overlaps were now suppressed. The reference of this study above is included in our manuscript (reference number 19) in the Methods section (Study design).

Also from
Answer: I could find only one overlap in the Discussion section and it was suppressed. Please let me know if there is more overlaps from this Systematic Review in our manuscript.

While we understand that this is work that you have previously published, and some of the same ideas are contained in these publications, please be aware that we cannot condone the use of text from previously published work.
While we understand that you may wish to express some of the same ideas contained in this publication, please be aware that we cannot condone the use of text from previously published work.
If there is overlap in the Methods section, please ensure to summarize the methods and cite the source. Please re-phrase these sections to minimise overlap.
Answer: We appreciate your attention and care with our manuscript. We hope that we have addressed your concern.

-- Many thanks for confirming consent to participate was in written form. Please also copy/place it under the 'Ethics approval and consent to participate' sub-section in the Declarations.
Answer: It was included in the Declarations. Thank you.